Author's response to reviews

Title: Essential pre-treatment imaging examinations in patients with endoscopically-diagnosed early gastric cancer

Authors:

  Eri Horisoko (heri@trad.ocn.ne.jp)
  Yoshito Tsushima (tyoshito@showa.gunma-u.ac.jp)
  Ayako Takahashi-Taketomi (ayakorad@yahoo.co.jp)
  Mari Tokunaga (mari_tokunaga@galaxy.ocn.ne.jp)
  Keigo Endo (endo@med.gunma-u.ac.jp)

Version: 4 Date: 12 May 2010

Author's response to reviews: see over
May 12th, 2010

Dear Editor,

Thank you for accepting our manuscript. We all are very glad to hear that. We are addressing a revised manuscript entitled “Essential pre-treatment imaging examinations in patients with endoscopically-diagnosed early gastric cancer: 7476955092909178” and a letter giving a point-by-point response to the concerns.

Thank you for reviewing this paper.

Sincerely yours,
Eri Horisoko, MD
Major revisions (we require the author to make these changes)

Tables: tables should fit onto the width of an A4 portrait page.
If you cannot re-orientate the table to fit this size, please move it to the additional files. Tables in the additional files will be made available with the final manuscript upon publication but will not be included in the manuscript in situ. Please ensure that you update the text to cite additional files and include a legend in the additional files section.

I have corrected tables to fit onto the width of an A4 portrait page.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Highlighting/tracking:

I have removed all highlighting and tracking.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

I have checked my manuscript.